

PEER-REVIEW REPORT

Team Players in the Pathogenesis of MASLD: The Basis of Development of Pharmacotherapy

Scientific Quality: Grade B (Very good)

Language Quality: Grade B (Minor language polishing)

Conclusion: Accept (General priority)

Novelty of This Manuscript: Grade B (Good)

Creativity or Innovation of This Manuscript: Grade B (Good)

Scientific Significance of the Conclusion in This Manuscript: Grade B (Good)

Reviewer's comments

The review summarizes the factors of Metabolic dysfunctions associated steatotic liver disease very effectively. It appears that a detailed literature analysis has been made. The data presented is supported by beautiful illustrations. I think that the article will gather citations with its ability to summarize the subject effectively and will be beneficial to those interested in the subject.

Author's response:

Reviewer's did not identify any concerns to address.